Author's response to reviews

Title: Burnout among medical professors; a nationwide survey

Authors:

Joeri K. Tijdink (j.tijdink@vumc.nl)
Anton C.M. Vergouwen (a.vergouwen@slaz.nl)
Yvo M. Smulders (y.smulders@vumc.nl)

Version: 2 Date: 30 October 2013

Author's response to reviews: see over
Cover Letter II

Amsterdam, October 30th, 2013

Dear Editor,

On behalf of my co-authors, I am pleased to resubmit our manuscript “Burnout among medical professors: a nationwide survey” as an original research article to BMC Medical Education.

Thank you for your request. You requested some extra changes which we tried to accomplish in submitting the manuscript. Below we point out the changes made:

1. On the submission we included a title page
2. We discussed this briefly with one of the members of the ethical committee of our university medical centre and concluded that approval of an appropriate ethical committee was not needed to conduct this survey since we do not include patients nor patient material but only medical professors.
3. We changed the abstract and checked the criteria for abstracts. Does the abstract require specific information which is now missing?
4. We mention competing interests at the end of the manuscript
5. We made a proper document consisting of the authorship contributions statement.
6. We have no acknowledgements to mention.

Thank you very much for considering this resubmission. If any questions arise during the process, I would be happy to provide more information.

On behalf of all authors,

Yours faithfully,

J.K. Tijdink, MD, Psychiatrist
VU University Medical Centre
Department of Internal Medicine
PO Box 7057
1007 MB Amsterdam
The Netherlands

Email: j.tijdink@vumc.nl